

# An Innovative Influenza Vaccination Policy: Targeting Last Season's Patients

The Harvard community has made this article openly available. Please share how this access benefits you. Your story matters.

Citation	Yamin, Dan, Arieh Gavious, Eyal Solnik, Nadav Davidovitch, Ran D. Balicer, Alison P. Galvani, and Joseph S. Pliskin. 2014. "An Innovative Influenza Vaccination Policy: Targeting Last Season's Patients." PLoS Computational Biology 10 (5): e1003643. doi:10.1371/journal.pcbi.1003643. http://dx.doi.org/10.1371/journal.pcbi.1003643.
Published Version	doi:10.1371/journal.pcbi.1003643
Accessed	February 16, 2015 12:15:07 PM EST
Citable Link	http://nrs.harvard.edu/urn-3:HUL.InstRepos:12406883
Terms of Use	This article was downloaded from Harvard University's DASH repository, and is made available under the terms and conditions applicable to Other Posted Material, as set forth at http://nrs.harvard.edu/urn-3:HUL.InstRepos:dash.current.terms-of- use#LAA

(Article begins on next page)

# An Innovative Influenza Vaccination Policy: Targeting Last Season's Patients



# Dan Yamin<sup>1,2\*</sup>, Arieh Gavious<sup>2,3</sup>, Eyal Solnik<sup>2</sup>, Nadav Davidovitch<sup>4</sup>, Ran D. Balicer<sup>5</sup>, Alison P. Galvani<sup>1</sup>, Joseph S. Pliskin<sup>2,4,6</sup>

1 Department of Epidemiology of Microbial Diseases, Yale University, New Haven, Connecticut, United States of America, 2 Department of Industrial Engineering and Management, Ben Gurion University of the Negev, Beersheba, Israel, 3 Faculty of Business Administration, Ono Academic College, Kiryat Ono, Israel, 4 Department of Health Systems Management, Ben Gurion University of the Negev, Beersheba, Israel, 5 Clalit Research Institute, Clalit Health Services, Tel Aviv, Israel, 6 Department of Health Policy and Management, Harvard School of Public Health, Boston, Massachusetts, United States of America

# Abstract

Influenza vaccination is the primary approach to prevent influenza annually. WHO/CDC recommendations prioritize vaccinations mainly on the basis of age and co-morbidities, but have never considered influenza infection history of individuals for vaccination targeting. We evaluated such influenza vaccination policies through small-world contact networks simulations. Further, to verify our findings we analyzed, independently, large-scale empirical data of influenza diagnosis from the two largest Health Maintenance Organizations in Israel, together covering more than 74% of the Israeli population. These longitudinal individual-level data include about nine million cases of influenza diagnosed over a decade. Through contact network epidemiology simulations, we found that individuals previously infected with influenza have a disproportionate probability of being highly connected within networks and transmitting to others. Therefore, we showed that prioritizing those previously infected for vaccination would be more effective than a random vaccination policy in reducing infection. The effectiveness of such a policy is robust over a range of epidemiological assumptions, including crossreactivity between influenza strains conferring partial protection as high as 55%. Empirically, our analysis of the medical records confirms that in every age group, case definition for influenza, clinical diagnosis, and year tested, patients infected in the year prior had a substantially higher risk of becoming infected in the subsequent year. Accordingly, considering individual infection history in targeting and promoting influenza vaccination is predicted to be a highly effective supplement to the current policy. Our approach can also be generalized for other infectious disease, computer viruses, or ecological networks.

Citation: Yamin D, Gavious A, Solnik E, Davidovitch N, Balicer RD, et al. (2014) An Innovative Influenza Vaccination Policy: Targeting Last Season's Patients. PLoS Comput Biol 10(5): e1003643. doi:10.1371/journal.pcbi.1003643

Editor: Katia Koelle, Duke University, United States of America

Received November 4, 2013; Accepted April 9, 2014; Published May 22, 2014

**Copyright:** © 2014 Yamin et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Funding:** This study was partially funded by the Israeli National Institute for Health Policy Research (NIHP) as a part of project number 10/151A, the James B. Crystal/Lillian Yaros Philanthropic Fund, and NIH grant U01 GM105627. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

\* E-mail: dan.yamin@yale.edu

## Introduction

Influenza has a long history of causing substantial morbidity, mortality and economic losses annually [1–3]. In Israel, influenza is responsible for about 801,200 reported infections (around 10% of the population), 4130 hospitalizations, 1140 deaths, and economic costs of 261 million dollars [3,4], while in the US, influenza is responsible for 610,600 life years lost and economic loss of \$87.1 billion annually [1]. Influenza vaccination is the primary approach to reduce the disease burden and is important not only for those vaccinated, but also to reduce transmission [2]. Recommendations by the World Health Organizations (WHO) [5], the U.S. Center for Disease Control and Prevention (CDC) [2], as well as the Israeli Ministry of Health have prioritized vaccination based on age, profession, and co-morbidities. However, these recommendations have not considered individual influenza infection history as an indication of future risk that can be used to supplement current policies.

An individual's infection risk is governed by their contacts as manifested by their social interactions. A contact network model captures the patterns of interactions that expose individuals to potential transmission. In the context of contact network epidemiology, centrals, individuals characterized by higher connectivity than average, are more likely both to become infected and to transmit infection [6,7]. Thus, prioritizing the vaccination of centrals could be effective in curtailing influenza transmission by reducing the network connectivity. However, identifying centrals is challenging [8], because the contact network is generally unknown.

One study [6] offered a novel way to reach the centrals in a network by randomly choosing individuals and asking them to deliver a vaccination dose to one of their contacts, an approach known as the 'acquaintance immunization strategy', suggests an indirect way to locate the centrals. Although this approach is an effective way to curtail transmission in both computer and population networks, it would be challenging to implement such a policy in the case of influenza vaccination.

In the current study we offer a practical way to devise a vaccination policy using the simple logic of targeting previous influenza patients. We propose that even in the absence of

## **Author Summary**

WHO/CDC recommendations prioritize influenza vaccinations primarily on the basis of age co-morbidities, but have never considered targeting vaccination for individuals previously infected with influenza. An individual's infection risk is governed by his or her contacts as manifested by his or her social interactions. Thus, through contact network simulations that capture contact patterns, we show here that individuals previously infected with influenza have a disproportionate probability of being highly connected within networks and thus serve as the super-spreaders. Accordingly, targeting them is effective in curtailing transmission. In addition to social interaction, a variety of factors, including genetics, co-morbidities, demographics, and epidemiological characteristics, may affect the risk and severity of influenza infection. Regardless of whether individuals are predisposed to infection because of these factors, or social interactions, we show that they can be identified through previous infection. Empirically, our analysis of medical records of influenza diagnosed in both hospitals and clinics confirms that in every age group, case definition for influenza, clinical diagnosis, and year tested, patients infected in the year prior had a substantially higher risk of becoming infected in the subsequent year. Thus, considering individual infection history in targeting and promoting influenza vaccination is predicted to be a highly effective supplement to the current prioritizations as it focuses on people with greater risk to become infected and transmit.

information about network structure, centrals are at higher risk of influenza infection and can thus be identified as being disproportionately represented in the pool of individuals who were previously infected. Further, in addition to social interaction, a variety of factors, such as genetics, co-morbidities, demographics, and epidemiological characteristics [2], may affect the risk and severity of infection, and remain relatively invariable over time. Regardless of whether individuals are predisposed to infection because of these factors or contact connectedness, they can be identified through previous infection. This approach is more straightforward than attempting to target individuals based on all possible risk factors, particularly as some risk factors may be unknown, difficult to identify or politically challenging to implement. Although the effectiveness of a policy that targets those previously infected initially seems to be counter-intuitive, since previously infected are likely to have partial protection against subsequent infection due to cross-reactivate antibodies [9], we found that previously infected individuals are much more likely to be infected even when taking into account biologically realistic rates of cross-reactivity [10].

Our findings are based on contact network epidemiology simulations and confirmed by empirical clinical data provided by the two largest Health Maintenance Organizations (HMOs) in Israel, covering more than 74% of the Israeli population. Our study is the first to address the interplay among previous infection history, immunological cross-reactivity, and social behavior as the basis for an innovative yet feasible supplement to the current influenza vaccination policies.

## Methods

#### Ethics statement

The surveillance data were analyzed anonymously, and approved to be used by the Clalit health services sub-Helsinky institutional review board, signed and approved by Dr. Eitan Wertheim, protocol number 127/2012.

#### Contact network simulations

Our simulations were applied to an epidemiological contact network based on the Portland population [11]. The Portland contact network derives from detailed microscopic simulationbased modeling and integration techniques performed by the Network Dynamic and Simulation Science Laboratory (NDSSL) at Virginia Tech with the purpose of creating a contact network reflecting an urban population [11]. The network includes 1,575,861 nodes, each of which represent an individual and 19,681,820 edges, each of which represents a contact between individuals.

To determine the robustness of our results, we validated our findings on three alternative small-world [12] scale-free networks: the Brightkite location-based network, the Gowalla location-based network, and the Barabási algorithm based network [13-15]. These networks vary in terms of the number of contacts, clustering coefficients, and the node-to-node distance [12]. The Brightkite location-based network is based on service providers where users share their locations by checking-in. The network was generated by the Stanford Network Analysis Project using their public Application Programming Interface, which consists of 58,228 nodes and 214,078 edges [16]. The Gowalla location-based contact network is a website where users share their locations by checking-in. The network was collected by the Stanford Network Analysis. It consists of 196,591 nodes and 950,327 edges [13]. We also created a network with 100,000 nodes and 400,000 edges according to the Barabási algorithm [14].

To evaluate centrality for each node in the networks, we calculated two common measures: number of contacts and K-shell decomposition values (K-shell) (Figure S1). Compared with the straightforward number of contacts measure, K-shell also take into account the global connectivity of the nodes to which a node is connected [17,18].

We used the Susceptible-Infectious-Recovered (SIR) compartmental model [19] to evaluate disease spread within the networks. According to each network configuration, an individual may infect only susceptible neighbors (i.e., nodes with whom they have edges). Given that not all individuals will be susceptible in the beginning of each season [10,20,21], we parameterized transmissibility using the effective reproductive number,  $R_e$  [21–23], defined as the average number of secondary infections resulting from each infective person [19] (Table 1). Protection following infection may last longer than a year [10] due to cross-reactivity between years. We considered the entire possible range of cross-reactivity from 0 to 100%, where 0% corresponds to no immunological protection and 100% corresponds to full protection acquired from influenza infection in the prior year. Depending on vaccine efficacy (Table 1), we assume that a proportion of individuals who are vaccinated are protected for one season [24].

We ran over one million simulations drawing parameter values from distributions that span a biologically realistic range (Table 1), as well as different vaccination rates and efficacies (Text S1). To determine whether previously infected individuals are more likely to be centrals, we evaluated the decile of the centrality score for each node in the network (based on the two measurements of centrality) and the risk ratio of becoming infected for a range of reproductive ratios and for different levels of cross-reactivity compared with the risk of a random individual. In each iteration of the simulation, we ran two successive influenza seasons. In the first season, we randomly vaccinated 0–40% of the population. In the second season of each simulation, we considered three policies: a Table 1. Parameter ranges and values for numerical simulations.

Symbol	Definition	Distribution/range checked	References
i <sub>o</sub>	Initial infection fraction	Uniform(0.0001,0.001)	[23]
R <sub>e</sub>	Effective reproductive ratio	1.2–1.6	[23,26,31]
V	Vaccination rate	0–0.4	[35]
R	Vaccination efficacy	0.5–0.8	[2]
D	Infection duration (in days)	Normal(3.8, 2)	[22,36]
θ	Cross-reactivity rate	0–1	
Т	Daily susceptibility rate between two neighbors	Uniform (0.012,0.087)	Supplements

doi:10.1371/journal.pcbi.1003643.t001

Random Vaccination Policy (RVP), an acquaintance immunization policy (AIP), which targets the acquaintances of a random node [6], and a vaccination policy prioritizing those who were infected in the previous season, Previously Infected Policy (PIP). For the *PIP* strategy, we assumed previously infected individuals are vaccinated first. If any vaccine doses remain after those prioritized have been vaccinated, the remaining doses are randomly distributed to the rest of the population.

#### Empirical demonstration

Our primary data were provided by Clalit Health Services and Maccabi Health Services and included demographic, socioeconomic, and ethnic data [25]. Clalit is the largest HMO in Israel, with membership varying between 3.47–3.72 million, and constituting about 53% of the Israeli population during the 2003–2012 the study period.

The Clalit dataset includes codes of the diagnosis recorded by the physicians in clinics according to the ICD9 protocol (codes 487 for 'influenza' and 465 for 'acute upper respiratory infections of multiple or unspecified sites') as well as hospitalizations due to influenza or pneumonia (ICD9, 486 'pneumonia organism unspecified' as well as codes 487, 465). Maccabi is the second largest HMO in Israel with 1.4–1.8 million members, constituting about 23% of the Israeli population, during the study period. Their records include full datasets of 12 years from September 1998 to April 2010 with 380,000 records. The data included influenza cases diagnosed in clinics according to the ICD10 protocol (code J11). Our case definition for influenza-like-illness (ILI) in both of the health maintenance organizations is detailed in Text S2.

As influenza infection rates depend on age, we stratified our data analysis by: 0–5, 6–15, 16–25, 26–35, 36–49, 50+. This division also facilitates evaluating the age-specific prioritization of the recommendations of the U.S. CDC as well as the Israeli Health Ministry which currently focus on ages 0–5 and individuals above age 50. In addition, we considered the relative risk of age group 25–35 which, along with children, are disproportionately responsible for transmission [26].

Rather than a case of influenza, an ILI infection might indicate elevated risk for influenza, because transmission routes of many upper respiratory diseases are similar. Thus, an ILI might serve as a predictor of elevated risk for both ILI and actual influenza. For each age group, in both HMOs, we calculated the relative risk of infection for those previously diagnosed with ILI compared with others in the same age group that had not been diagnosed in prior season.

In the Clalit dataset we stratified our ten seasons of data into eleven pairs of two consecutive seasons and calculated the risk of outpatient influenza in season i for influenza outpatient patients diagnosed in season i-1, the risk of outpatient influenza in season ifor patients hospitalized in season i-1, the risk of becoming hospitalized with influenza in season i for outpatient patients diagnosed in season i-1, and the risk of becoming hospitalized with influenza in season i for patients hospitalized in season i-1. In the Maccabi dataset, we stratified our twelve seasons of data into eleven pairs of two consecutive seasons and calculated the risk of outpatient influenza in season i for influenza outpatient patients diagnosed in season i-1 (Text S2).

Not all influenza patients seek medical treatment [1,27], and some people might have higher tendency to seek medical treatment when infected with influenza than others, potentially leading to an overestimation relative risks. In addition, individuals are likely to visit the same physician when infected with influenza, and the latter might not diagnosis the infection as ILI. Thus, we compared evaluations of the policies under the conservative assumption that individuals can be divided into those that either seek medical treatment when infected, or not seek medical treatment. Under this conservative assumption, we calculated an adjusted relative risk of outpatient infection by removing in each age group members who were never an outpatient along the entire period tested.

# Results

The risk of future influenza infections for individuals is determined by the interplay between two countering factors: social interaction, which governs exposure probability, and crossreactivity protection acquired from previous infection. We found that these factors will affect: 1) the conditions under which central individuals have a higher risk of infection than non-centrals, 2) the conditions under which individuals infected in the prior season have a higher risk to become infected in a succeeding season, and, consequently, 3) the conditions in which the targeting of last season's patients for influenza vaccination will serve as an effective policy to decrease morbidity. Overall, our analysis demonstrates that social interaction dominated in determining the overall effectiveness of targeting influenza vaccination to previously infected individuals in vast ranges of realistic conditions, as elaborated below.

Our contact network simulations reveal that individuals infected with influenza in the prior year have a greater probability of being more central than others. This finding is robust to the two different measurements of centrality considered (Figures 1 and S2), and over the range of biologically realistic epidemiological parameters for influenza (Table 1). The results were more sensitive to the initial proportion of susceptibles. Specifically, the greater the crossreactivity and attack rate are in the previous season, the lower is



**Figure 1. The relative risk of infection given parameters of centrality.** The mean and 95% confidence interval of relative risk of infection for an individual compared to the rest of the population, given his/her K-shell (panels A and B), and number of contacts (panels C and D) for cross-reactivity levels of 0% (panels A and C) and 80% (panels B and D) for effective reproductive number,  $R_e = 1.2$  (dotted red), 1.4 (dashed blue) and 1.6 (dot-dashed green). A relative risk above one represents higher risk of infection, compared with the rest of the population. doi:10.1371/journal.pcbi.1003643.g001

the risk of infection for the central individuals relative to the rest of the network. For example, in the Portland network, we found that individuals with higher K-shell values or higher number of contacts than others have a higher risk of infection when cross-reactivity ranges between 0–60%. For cross-reactivity higher than 60%, and particularly for the upper estimates of influenza transmissibility (i.e.,  $R_e \ge 1.4$ ), individuals in the ninth or tenth decile with the highest Kshell value or number of contacts are less likely to get infected relative to individuals in the fifth decile, because the former were more likely to have cross-protection from previous exposure. Nevertheless, even in this case, individuals who have a centrality score above the median have a higher risk for re-infection than individuals below the median (Figures 1 and 2). Collectively, centrals are disproportionately represented in those who were previously infected even when cross-reactivity reached 80%.

For seasons between which there is no cross-reactivity, we showed analytically that prior influenza patients have a higher risk of infection in the succeeding season (text S1). Consistent with this analytical finding, our simulations on the Portland network also demonstrate that when there is no cross-reactivity, the relative risk of previous patients ranges from 2.3–3.1 compared to individuals not infected in the prior season. However, this risk can also fall below one in cross-reactivity rates above 60% (Figure S3).

We compared three policies: a Random Vaccination Policy (*RVP*), the acquaintance immunization policy (*AIP*), and a vaccination policy prioritizing those who were infected in the previous season, Previously Infected Policy (*PIP*) (see methods). Overall, in the Portland network, results demonstrate that *PIP* is more effective than *AIP* in reducing morbidity rates even in cases where cross-reactivity is as high as 36-72%. *PIP* is also more preferable than *RVP* even in cases where cross-reactivity is as high as 57-78% (Figures 3, Figure S4 and Figure S5). For example, when cross-reactivity of 40% has been observed and 15% are vaccinated the mean risk of infection would be 14%, 11.5% and 5.6% for *RVP*, *AIP* and *PIP*, respectively; whereas when 30% are vaccinated the mean risk would be 8.6%, 4.6%, and 3.6% for *RVP*, *AIP* and *PIP*, respectively.

The variability observed in our simulations (Figure 3, Figure S4 and Figure S5) arises from the sensitivity of *PIP* to two drivers. The first driver is the vaccination coverage in the previous season or infected in the previous season and who remained protected due to cross-reactivity as those individuals cannot be detected by our policy. The second driver is the incidence in the previous season. If the prior incidence is relatively low, all of those who had been infected in the prior year would be vaccinated in the subsequent year, but the remaining doses will be randomly distributed to the



**Figure 2. Mean risk of infection following vaccination.** The mean risk of infection evaluated over the parameters ranges in Table 1 for *RVP* (dashed blue), *AIP* (dashed-doted red), *PIP* (dashed green), as well as no vaccination (solid black), for cross-reactivity levels of A and B) 0%, C and D) 40%, E and F) 60% G and H) 80%. In the second season, for *RVP*, *AIP* and *PIP* strategies, vaccination coverage for A, C, E and G) 15% and for B, D, F and H) 30%, and vaccine efficacy of 75%. *PIP* is preferable than *RVP* in reducing morbidity for panels A–F, and more preferable than *AIP* for panels A–D. As explained in the main text, the risk of infection decreases as the cross-reactivity increases. doi:10.1371/journal.pcbi.1003643.g002

rest of the population, thereby making our policy more similar to *RVP*.

To verify the robustness of our results, we determined the effectiveness of *PIP* for four networks across a range of biologically realistic epidemiological parameters for influenza (Table 1). *PIP* was found to be more effective than *RVP* when cross-reactivity was as high as 57-75% in the Portland Network, 55-80% in the Barabási algorithm-based network, 80-95% in the Brightkite Network, and 75-90% in the Gowalla Network. In comparison with *AIP*, *PIP* was effective in lower values of cross-reactivity levels,

ranged from 35–72.5% in Portland Network, 0–40% Brightkite Network, and 0–40% in the Gowalla Network (Figure 3). *PIP* was less effective than *AIP* in the Barabási algorithm based network, possibly as a result of the homogeneity of K-shell values among nodes in this network (Figures S1).

### Empirical demonstration

Given that influenza attack rate ranges epidemiologically between 5–15% [2], our case definition for influenza may be under-reported in the Maccabi dataset and over-reported in the



**Figure 3. Mean indifference curves for** *PIP* **vs.** *RVP* **and** *PIP* **vs.** *AIP*. The curves are shown as a function of the effective reproductive number and cross-reactivity for A) the Portland Network B) Barabási algorithm-based network C) Brightkite Network D) Gowalla Network. Above each curve, *RVP/AIP* is the recommended policy, whereas below the curve, *PIP* is recommended. doi:10.1371/journal.pcbi.1003643.g003

Clalit dataset. Nevertheless, our analysis of over nine million medical records showed that patients diagnosed with ILI in the previous season have a substantially higher risk to be diagnosed with ILI in the succeeding year (Table 2). Excepting children below five who had a lower risk of being hospitalized, diagnosed with outpatient influenza in the season prior, our results were robust for both HMOs, for each age group and for almost every year tested. For example, in the Maccabi dataset within the age group of 35-49, only 1.45% of the population was outpatient ILI. However, for individuals within this age group who were diagnosed in the previous year, the risk of becoming infected in the subsequent year was 11.35%. In the Clalit dataset, within the age group 35-49, 9.36% were outpatient ILI. The risk was about four times higher in patients who were outpatient ILI, and two fold in patients previously hospitalized compared with patients who did not seek medical treatment of ILI in the year prior (Table 2).

Even when we calculated an adjusted relative risk of outpatient influenza by considering only members who were diagnosed with outpatient influenza at least once in the study period (i.e. 2003– 2012 in Clalit dataset and 1998–2010 in Maccabi dataset), the relative risk for individuals previously outpatient was still higher than one in each age group (Table 2). This finding demonstrates that, in addition to age, an individual's infection history plays an important role in determining their subsequent risk of infection.

# Discussion

Our work shows that considering individual infection history in targeting and promoting influenza vaccination would be an effective supplement to the current policy which prioritizes individuals on the basis of age and co-morbidities. Our findings highlight the fundamental role that an individual's social behavior plays in disease transmission [28], and reveals that in the interplay between cross-reactivity and individual risk factors the latter dominates in determining the overall effectiveness of targeting influenza vaccination to individuals infected in the prior season.

Our simulations demonstrate that targeting individuals previously diagnosed with influenza can be effective even if crossreactivity is as high as 55–80%. Empirical studies suggest that cross-reactivity for a specific type of influenza is typically below or within this range [10]. Additionally, there can be two or three sub-types of influenza circulating within a single season [29] with dominance shifting among sub-types between successive years.

Surveillance systems monitor reporting rates rather than actual prevalence, and thus also include misdiagnoses. Nonetheless, ILI diagnoses, even if from a different etiology, might indicate elevated risk of future infection with influenza, because transmission routes and associated risk factors of many upper respiratory diseases are similar. Given that respiratory infections other than influenza will not elicit cross-reactive antibodies, these misdiagnosed individuals may be at even higher risk for future influenza infection than those who were previously infected with influenza. To evaluate the increased risk of those who were previously infected with influenza versus other respiratory infections, future research should stratify between clinical diagnosis and laboratory confirmation.

While social interactions modeled in our contact network simulations are fundamental to influenza transmission, other factors including genetics, co-morbidities, and demography, contribute to determine risk for an individual. Similar to social tendencies, these other factors will also remain relatively invariable for an individual from year to year. Consequently, previous, current, and future infection risk can be even more effectively predicted by prior infection than what we conservatively estimated from social interaction alone.

If *PIP* is implemented every influenza season, individuals with high connectivity might be targeted in the first year, and therefore will reduce their risk of infection in the subsequent year. However,

s tested.
year
the
over
group
age
and
history
infection
by
stratified
of infection
isk o
relative r
and
risk
Means
'n
Table

Health maintenance organization	Clinical outcome	Population by age	0-4	5-14	15-24	25-34	35-49	>50
<b>Clalit Health Services</b>	Outpatient visits	Risk for entire population	26.36%	11.5%	9.48%	10.44%	9.36%	7.91%
		Risk given outpatient last year	44.2%	31.09%	26.91%	27.26%	28.83%	24.89%
		Risk given hospitalized last year	29.41%	16.24%	14.27%	14.02%	12.77%	7.38%
		Relative risk given outpatient last year	2.32 (1.94–2.62)	3.62 (2.90–4.02)	3.59 (3.03–3.89)	3.36 (3.07–3.69)	4.07 (3.71–4.52)	3.93 (3.59–4.45)
		Relative risk given hospitalized last year	1.44 (1.25–1.62)	1.76 (1.43–2.3)	1.86 (1.57–2.10)	1.60 (1.39–1.80)	1.93 (1.57–2.25)	1.12 (0.90–1.27)
		Adjusted relative risk given outpatient last year	2.21 (1.91–2.49)	2.36 (2.06–2.61)	2.10 (1.89–2.25)	2.07 (1.88–2.29)	2.24 (2.08–2.52)	1.98 (1.75–2.26)
		Adjusted relative risk given hospitalized last year	1.37 (1.22–1.56)	1.15 (1.02–1.56)	1.09 (0.91–1.22)	1.01 (0.85–1.21)	1.06 (0.90–1.21)	0.56 (0.48–0.62)
	Hospitalizations	Risk for entire population	0.39%	0.06%	0.03%	0.04%	0.07%	0.32%
		Risk given outpatient last year	0.31%	0.08%	0.13%	0.05%	0.09%	0.45%
		Risk given hospitalized last year	3.22%	3.60%	4.24%	3.23%	3.85%	7.21%
		Relative risk given outpatient last year	0.71 (0.53–0.77)	1.33 (1.02–1.64)	7.04 (4.01–12)	1.14 (0.93–1.41)	1.42 (1.17–1.73)	1.42 (1.31–1.54)
		Relative risk given hospitalized last year	8.02 (6.58–10.50)	66.6 (47.9–84.8)	219.42 (76.4–564.6)	82.7 (62.1–97.9)	70.13 (42.5–114.4)	25.48 (22.68–31.05)
Maccabi Health Services	Outpatient visits	Risk for entire population	2.04%	2.01%	1.35%	1.58%	1.45%	0.93%
		Risk given outpatient last year	10.12%	9.02%	8.33%	10.28%	11.35%	13.77%
		Relative risk	5.69 (4.31–7.51)	5.13 (3.06–6.77)	7.09 (3.75–10.30)	7.47 (5.65–10.5)	8.81 (6.21–12.16)	17.11 (11.41–26.12)
		Adjusted relative risk	1.32 (0.96–1.68)	1.10 (0.84–1.46)	1.26 (0.89–1.83)	1.27 (0.94–1.74)	1.40 (1.17–1.92)	1.61 (1.48–2.46)
The annual minimum and r adjusted relative risk incluc  doi:10.1371/journal.pcbi.10(	maximum values observe ded as members only in 03643.t002	ed in the years are presented in parentheses. T dividuals who were diagnosed as influenza p	he relative risk was ca atients at least once a	lculated compared v slong the entire peri	vith individuals from the iod evaluated (2003–201	same HMO that soug 2 in Clalit dataset an	ht medical treatmen d 1998–2010 in Mace	: in the year prior. Th abi dataset).

as a result of this reduced risk, in the third year from initiating the *PIP* policy, they will be less likely to be infected and subsequently targeted. Thus, future studies could evaluate the marginal benefit of considering infection and vaccination history of individuals over several seasons relative to the prior season alone in determining vaccine targets.

Targeting previously infected individuals is a relatively straightforward approach to implement in an HMO system with electronic medical records. For instances, individuals previously infected could be flagged within the electronic records for vaccination targeting by mailing pamphlets, telephone reminders or physician recommendations, practices shown to be effective in promoting influenza vaccination [2]. The suggested policy could reach several sub-populations, such as those based on socioeconomic status and ethnicity, that correlated with vaccine uptake and infection rates [30], but which have not been prioritized in the past due to ethical or political reasons. Furthermore, a high level of public adherence to this targeted strategy is likely to be achievable, given that individuals who were recently ill with influenza will probably be responsive to strategies that reduce their personal risk, as has been shown to be a primary motivator in vaccination decisions [23,31,32].

We demonstrated the potential benefit of targeting last season's patients for influenza, but such policy may also be applicable to other diseases including respiratory syncytial virus, pneumococcal infections and malaria, for which re-infection is common. Our approach may be generalized to networks outside the public health field, such as ecology and computer science. For example, our approach may determine which computers should be prioritized for antivirus software installations. In fact, a previous study on computer networks showed that computers that were attacked in one simulation run are most prone to attack in other simulation runs [33]. The authors even suggested little variation in the number of reinfections experienced by the same computer in different simulation studies, making our approach likely to be highly effective. In another example, our approach may also be helpful in ecological networks to identify and invest efforts to protect species with most essential to community stability [34].

In summary, we modeled the interplay among previous infection history, immunological cross-reactivity, and social behavior as the basis to generate an innovative influenza vaccination policy. Through contact network simulations we showed that individuals infected in the year prior have higher connectivity in the network, and subsequently increased risk of infection and transmission. Empirically, our analysis of the medical records confirms that in every age group, case definition for influenza, clinical diagnosis and year tested, patients infected in the year prior had a substantially higher risk of becoming infected in the subsequent year. Accordingly, the targeting of individuals infected in the prior year is predicted to be a highly effective supplement to the current policy.

# **Supporting Information**

**Figure S1 Distribution of measurements of centrality.** The distribution of 1) Number of contacts, and 2) K-shell is shown

#### References

for A) the Portland Network, B) Brightkite Network, C) Gowalla Network, D) Barabási Algorithm-Based Network. (TIF)

Figure S2 Relative risk of infection given parameters of centrality. The mean and 95% confidence interval of the relative risk of infection for an individual compared to the rest of the population, given his/her K-shell (panels A and B), and number of contacts (panels C and D) for cross-reactivity levels of A) 20% B) 40% C) 60% for effective reproductive number,  $R_e = 1.2$  (dotted red), 1.4 (dashed blue) and 1.6 (dot-dashed green). A relative risk above one represents higher risk of infection, compared with the rest of the population. The figure complements Figure 1. (TIF)

Figure S3 Effects of previous illness on future infection proportion. The mean and 95% confidence interval of the relative risk of infection in individuals previously infected versus individuals not previously infected depending on  $R_e$  and cross-reactivity. The black line represents relative risk equal to one. (TIF)

**Figure S4 Boxplot risk of infection following vaccina-tion.** Box-and-Whisker plots of the difference between the risk of infections for *RVP* and *PIP*, and *AIP* and *PIP*, over the parameters ranges in Table 1, for cross-reactivity of A) 0%, B) 40%, C) 60%, and D) 80%, assuming vaccination coverage of 15% and vaccine efficacy of 75%. This figure corresponds to Figure 2 panels A, C, E, and F.

(TIF)

**Figure S5 Boxplot risk of infection following vaccina-tion.** Box-and-Whisker plots of the difference between the risk of infections for *RVP* and *PIP*, and *AIP* and *PIP*, over the parameters ranges in Table 1, for cross-reactivity of A) 0%, B) 40%, C) 60%, and D) 80%, assuming vaccination coverage of 30% and vaccine efficacy of 75%. This figure corresponds to Figure 2 panels B, D, F, and G.

(TIF)

**Figure S6 Frequency of ILI diagnosis.** The right axis refers to influenza and pneumonia diagnosed in hospitals. The left axis refers to influenza diagnosis.

(TIF)

# **Text S1 Supporting information for contact network analysis.** (DOCX)

**Text S2** Supporting information for data analysis. (DOCX)

#### Acknowledgments

The authors would like to thank Dr. Chad Wells, Dr. Michal Brumberg, Dr. Varda Shalev, Dr. Moshe Hoshen, Karin Daly, Hilla Burstein and Rita Umile.

#### **Author Contributions**

Conceived and designed the experiments: DY AG ES. Performed the experiments: DY ES. Analyzed the data: DY ES. Contributed reagents/ materials/analysis tools: RDB ND JSP APG AG. Wrote the paper: DY APG JSP AG.

 Fiore AE, Uyeki TM, Broder K, Finelli GL, Singleton J, et al. (2010) Prevention and Control of Influenza with Vaccines. Available: http://www. cdc.gov/mmwr/preview/mmwrhtml/rr5908a1.htm. Accessed 30 October 2012.

Molinari N-AM, Ortega-Sanchez IR, Messonnier ML, Thompson WW, Wortley PM, et al. (2007) The annual impact of seasonal influenza in the US: measuring disease burden and costs. Vaccine 25: 5086–5096. Available: http:// dx.doi.org/10.1016/j.vaccine.2007.03.046. Accessed 5 March 2013.

- Stephenson I, Hayden F, Osterhaus A, Howard W, Pervikov Y, et al. (2010) Report of the fourth meeting on "Influenza vaccines that induce broad spectrum and long-lasting immune responses", World Health Organization and Wellcome Trust, London, United Kingdom, 9–10 November 2009. Vaccine 28: 3875– 3882. Available: http://www.ncbi.nlm.nih.gov/pubmed/20398616. Accessed 18 November 2012.
- Balicer RD, Huerta M, Davidovitch N, Grotto I (2005) Cost-benefit of stockpiling drugs for influenza pandemic. Emerg Infect Dis 11: 1280– 1282. Available: http://www.pubmedcentral.nih.gov/articlerender.fcgi?artid = 3320484&tool = pmcentrez&rendertype = abstract. Accessed 2 January 2014.
- WHO | Influenza (2003). World Heal Organ. Available: http://www.who.int/ mediacentre/factsheets/2003/fs211/en/. Accessed 1 January 2013.
- Cohen R, Havlin S, ben-Avraham D (2003) Efficient Immunization Strategies for Computer Networks and Populations. Phys Rev Lett 91: 2–5. Available: http://link.aps.org/doi/10.1103/PhysRevLett.91.247901. Accessed 6 November 2012.
- Christakis N a, Fowler JH (2010) Social network sensors for early detection of contagious outbreaks. PLoS One 5: e12948. Available: http://www. pubmedcentral.nih.gov/articlerender.fcgi?artid = 2939797&tool = pmcentrez &rendertype = abstract. Accessed 7 November 2012.
- Galvani AP, May RM (2005) Dimensions of superspreading. Nature 438: 293– 295.
- Hancock K, Veguilla V, Lu X, Zhong W, Butler EN, et al. (2009) Cross-reactive antibody responses to the 2009 pandemic H1N1 influenza virus. N Engl J Med 361: 1945–1952. Available: http://www.ncbi.nlm.nih.gov/pubmed/19745214.
- Ikonen N, Strengell M, Kinnunen L, Osterlund P, Pirhonen J, et al. (2010) High frequency of cross-reacting antibodies against 2009 pandemic influenza A(H1N1) virus among the elderly in Finland. Euro Surveill 15: 1–8. Available: http://www.ncbi.nlm.nih.gov/pubmed/20144443.
- Data Products for Societal Infrastructures and Proto-Populations: Data Set 1.0. (n.d.). Virginia Polytech Inst State Univ. Available: ndssl.vbi.vt.edu/ Publications/ndssl-tr-06-006.pdf.
- Watts DJ, Strogatz SH (1998) Collective dynamics of "small-world" networks. Nature 393: 440–442. Available: http://www.nature.com/nature/journal/ v393/n6684/pdf/393440a0.pdf. Accessed 5 November 2012.
- Cho E, Myers SA, Leskovec J (2011) Friendship and Mobility: User Movement In Location-Based Social Networks. KDD'11.
- Barabasi A-L (2002) Linked: The New Science of Networks. Perseus Books. Available: http://cs.stanford.edu/people/jure/pubs/mobile-kdd11.pdf. Accessed 26 December 2012.
- Synthetic Data Products for Societal Infrastructures and Proto-Populations: Data Set 1.0 (n.d.). Available: http://ndssl.vbi.vt.edu/Publications/ndssl-tr-06-006.pdf. Accessed 26 December 2012.
- SNAP: Network datasets: Brightkite (n.d.). Available: http://snap.stanford.edu/ data/loc-brightkite.html. Accessed 26 December 2012.
- Kitsak M, Gallos LK, Havlin S, Liljeros F, Muchnik L, et al. (2010) Identification of influential spreaders in complex networks. Nat Phys 6: 888– 893. Available: http://dx.doi.org/10.1038/nphys1746. Accessed 7 November 2012.
- Seidman SB (1983) Network structure and minimum degree. Soc Networks 5: 269–287. Available: http://dx.doi.org/10.1016/0378-8733(83)90028-X. Accessed 21 November 2012.
- Vynnycky E, White R (2010) An Introduction to Infectious Disease Modelling. Oxford University Press, USA.
- Katriel G, Stone L (2010) Pandemic dynamics and the breakdown of herd immunity. PLoS One 5: e9565. Available: http://www.pubmedcentral.nih.gov/ articlerender.fcgi?artid = 2837721&tool = pmcentrez&rendertype = abstract. Accessed 30 October 2012.
- 21. Huppert A, Barnea O, Katriel G, Yaari R, Roll U, et al. (2012) Modeling and Statistical Analysis of the Spatio-Temporal Patterns of Seasonal Influenza in

Israel. PLoS One 7: e45107. Available: http://dx.plos.org/10.1371/journal. pone.0045107. Accessed 9 October 2012.

- Fraser C, Donnelly CA, Cauchemez S, Hanage WP, Van Kerkhove MD, et al. (2009) Pandemic potential of a strain of influenza A (H1N1): early findings. Science 324: 1557–1561. Available: http://www.sciencemag.org/content/324/ 5934/1557.abstract. Accessed 30 October 2012.
- 23. Galvani AP, Reluga TC, Chapman GB (2007) Long-standing influenza vaccination policy is in accord with individual self-interest but not with the utilitarian optimum. Proc Natl Acad Sci U S A 104: 5692–5697. Available: http://www.pubmedcentral.nih.gov/articlerender.fcgi?artid = 1838447&tool = pmcentrez&rendertype = abstract.
- 24. Grohskopf L, Shay DK, Shimabukuro T, Sokolow L, Keitel W, et al. (2013) Prevention and Control of Seasonal Influenza with Vaccines: Recommendations of the Advisory Committee on Immunization Practices — United States, 2013– 2014. Available: http://www.cdc.gov/mmwr/preview/mmwrhtml/rr6207a1. httm?s\_cid = rr6207a1\_w. Accessed 5 January 2014.
- Memebership in HMO's (n.d.). Natl Insur Inst Isr. Available: http://www.btl. gov.il/Publications/survey/Pages/saker233.aspx.
- Medlock J, Galvani AP (2009) Optimizing influenza vaccine distribution. Science (80-) 325: 1705–1708. Available: http://www.sciencemag.org/content/ 325/5948/1705.abstract. Accessed 30 October 2012.
- Bridges CB, Thompson WW, Meltzer MI, Reeve GR, Talamonti WJ, et al. (2000) Effectiveness and Cost-Benefit of Influenza Vaccination of Healthy Working Adults. JAMA 284: 1655. Available: http://jama.jamanetwork.com/ article.aspx?articleid = 193139. Accessed 28 February 2014.
- Funk S, Salathé M, Jansen VAA (2010) Modelling the influence of human behaviour on the spread of infectious diseases: a review. J R Soc Interface 7: 1247–1256. Available: http://rsif.royalsocietypublishing.org/content/7/50/ 1247.short. Accessed 26 October 2012.
- CDC Seasonal Influenza (Flu) Past Weekly Surveillance Reports (n.d.). Available: http://www.cdc.gov/flu/weekly/pastreports.htm. Accessed 13 September 2013.
- Armstrong K, Berlin M, Schwartz JS, Propert K, Ubel P a (2001) Barriers to influenza immunization in a low-income urban population. Am J Prev Med 20: 21–25. Available: http://www.ncbi.nlm.nih.gov/pubmed/11137770.
- Yamin D, Gavious A (2013) Incentives' Effect in Influenza Vaccination. Manage Sci 59. Available: http://papers.ssrn.com/abstract=1967071. Accessed 16 November 2012.
- Brewer NT, Chapman GB, Gibbons FX, Gerrard M, McCaul KD, et al. (2007) Meta-analysis of the relationship between risk perception and health behavior: the example of vaccination. Health Psychol 26: 136–145. Available: http://www.ncbi.nlm.nih.gov/pubmed/17385964. Accessed 16 October 2012.
- 33. Knight JC, Elder MC (2000) On computer viral infection and the effect of immunization. Proceedings 16th Annual Computer Security Applications Conference (ACSAC'00). IEEE Comput. Soc. pp. 246–256. Available: http:// ieeexplore.ieee.org/lpdocs/epic03/wrapper.htm?arnumber = 898879. Accessed 18 February 2014.
- Solé R V, Montoya JM (2001) Complexity and fragility in ecological networks. Proc Biol Sci 268: 2039–2045. Available: http://rspb.royalsocietypublishing. org/content/268/1480/2039.short. Accessed 31 January 2014.
- Blank PR, Schwenkglenks M, Szucs TD (2009) Vaccination coverage rates in eleven European countries during two consecutive influenza seasons. J Infect 58: 446–458. Available: http://www.ncbi.nlm.nih.gov/pubmed/19446340. Accessed 30 October 2012.
- Cauchemez S, Carrat F, Viboud C, Valleron AJ, Boëlle PY (2004) A Bayesian MCMC approach to study transmission of influenza: application to household longitudinal data. Stat Med 23: 3469–3487. Available: http://www.ncbi.nlm. nih.gov/pubmed/15505892. Accessed 2 November 2012.